

Citation: Langston H, Fortes Francisco A, Doidge C, Roberts CH, Khan AA, Jayawardhana S, et al. (2024) Dynamics of *Trypanosoma cruzi* infection in hamsters and novel association with progressive motor dysfunction. PLoS Negl Trop Dis 18(6): e0012278. https://doi.org/10.1371/journal. pntd.0012278

Editor: Alain Debrabant, US Food and Drug Administration, UNITED STATES

Received: November 13, 2023

Accepted: June 7, 2024

Published: June 21, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0012278

Copyright: © 2024 Langston et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files.

RESEARCH ARTICLE

Dynamics of *Trypanosoma cruzi* infection in hamsters and novel association with progressive motor dysfunction

Harry Langston^{1¤a}, Amanda Fortes Francisco¹, Ciaran Doidge^{1¤b}, Chrissy H. Roberts², Archie A. Khan¹, Shiromani Jayawardhana¹, Martin C. Taylor¹, John M. Kelly¹, Michael D. Lewis^{1^{ac}*}

1 Department of Infection Biology, London School of Hygiene and Tropical Medicine, London, United Kingdom, 2 Department of Clinical Research, London School of Hygiene and Tropical Medicine, London, United Kingdom

¤a Current address: Imperial College London, London, United Kingdom

¤b Current address: Barts Cancer Institute, Queen Mary University of London, London, United Kingdom
¤c Current address: Division of Biomedical Sciences, Warwick Medical School, University of Warwick, Coventry, United Kingdom

* michael.d.lewis@warwick.ac.uk

Abstract

Chagas disease is a zoonosis caused by the protozoan parasite Trypanosoma cruzi. Clinical outcomes range from long-term asymptomatic carriage to cardiac, digestive, neurological and composite presentations that can be fatal in both acute and chronic stages of the disease. Studies of T. cruzi in animal models, principally mice, have informed our understanding of the biological basis of this variability and its relationship to infection and host response dynamics. Hamsters have higher translational value for many human infectious diseases, but they have not been well developed as models of Chagas disease. We transposed a real-time bioluminescence imaging system for T. cruzi infection from mice into female Syrian hamsters (Mesocricetus auratus). This enabled us to study chronic tissue pathology in the context of spatiotemporal infection dynamics. Acute infections were widely disseminated, whereas chronic infections were almost entirely restricted to the skin and subcutaneous adipose tissue. Neither cardiac nor digestive tract disease were reproducible features of the model. Skeletal muscle had only sporadic parasitism in the chronic phase, but nevertheless displayed significant inflammation and fibrosis, features also seen in mouse models. Whereas mice had normal locomotion, all chronically infected hamsters developed hindlimb muscle hypertonia and a gait dysfunction resembling spastic diplegia. With further development, this model may therefore prove valuable in studies of peripheral nervous system involvement in Chagas disease.

Author summary

Chagas disease is a caused by American trypanosomes (*Trypanosoma cruzi*). These are microscopic parasites that circulate in wild mammals across most of the Americas and

Funding: The research was supported by UK Medical Research Council (https://www.ukri.org/ councils/mrc/) grants MR/T015969/1 to J.M.K. and MR/R021430/1 to M.D.L. No authors received salary directly from the funders. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

can also be transmitted to humans. Much of our knowledge about how *T. cruzi* causes Chagas disease comes from studies of infections in mice, but the data do not capture the full range of clinical outcomes seen in humans. For many other pathogens the hamster has proved to be a valuable model of human infections. We therefore aimed to apply some of the latest advances in *T. cruzi* infection imaging technology to studies in this alternative experimental model. In the early stages, parasites were widely disseminated throughout the body, but after several months parasites became almost entirely restricted to the skin. Hamsters did not show signs of heart or gut disease, which are common in humans, but they did develop skeletal muscle pathology. Stiffness in the hindlimbs grew progressively worse and resulted in a visibly altered gait, suggestive of damage to the nervous system. With further development, this model may therefore prove valuable in studies of peripheral nervous system involvement in Chagas disease.

Introduction

Chagas disease (American trypanosomiasis) is caused by infection with *Trypanosoma cruzi*, a protozoan parasite. Approximately 6 million people are infected and the disease causes ~10,000 deaths annually as well as a large morbidity burden in affected populations [1]. Clinical outcomes are highly heterogeneous, encompassing muscle and nervous tissue pathologies affecting the heart and gastro-intestinal tract, as well as long-term asymptomatic carriage [2] Pathogenesis is thought to be primarily a result of collateral damage to infected tissues from the host's cellular immune response, which leads to fibrosis, microvascular abnormalities, denervation and consequent organ dysfunction [3–5].

Animal infection models are critical to research on the biological basis of Chagas disease. *T. cruzi* is a very cosmopolitan parasite in its natural host range [6] and a variety of species have been used in experimental settings, including rats, rabbits, dogs, guinea pigs, opossums and non-human primates, often reproducing important features of the human disease spectrum [7]. Mice, however, are by far the most widely used species; in most cases they develop inflammatory cardiomyopathy [8–14] and a few models are suitable for studying aspects of digestive disease [15–18]. The compatibility of murine models with technologies such as bioluminescence imaging (BLI), which enable real-time monitoring of infections, is a major advantage and has led to significant advances in understanding how spatio-temporal infection dynamics relate to pathology development [9,15,19]. Mice have been less well developed as models of advanced digestive megasyndromes, peripheral neuropathy or CNS infection, an important cause of acute mortality in human patients [20,21]. Furthermore, the use of a few inbred mouse strains limits our ability to study the impact of host genetic diversity on disease pathogenesis and clinical outcome heterogeneity.

The aim of this study was to develop an *in vivo/ex vivo* BLI system for *T. cruzi* infection in the Syrian hamster, *Mesocricetus auratus*. Hamsters can show closer alignment with human diseases than mice for a variety of infectious agents, for example *Clostridium difficile*, SARS-CoV-2, West Nile Virus, adenoviruses, *Leishmania donovani* and *Schistosoma haematobium* [22–26]. Hamsters have also been described as useful models of Chagas cardiac disease [27–33] and in some cases are reported to develop a dilated large intestine indicative of megacolon [28]. They are also one of the few species readily available, besides mice, whose physical size is compatible with standard *in vivo* imaging chambers. We therefore reasoned that a hamster *T. cruzi* BLI system could add value to the landscape of experimental Chagas disease *in vivo* models.

Methods

Ethics statement

Animal work was approved by the London School of Hygiene and Tropical Medicine Animal Welfare and Ethical Review Board and carried out under UK Home Office project licence (PPL 70/8207) in accordance with the UK Animals (Scientific Procedures) Act.

Parasites, animals and infections

T. cruzi CL Brener (genetic lineage TcVI) constitutively expressing the red-shifted firefly luciferase gene *PpyRE9h* [34] were used in all experiments. Infectious trypomastigotes were generated as previously described [34]. Animals used were female CB17 SCID mice aged 8–10 weeks (bred in house), female BALB/c, C57BL6 and C3H/HeN mice aged 6–8 weeks (Charles River) and female Syrian hamsters aged 6–8 weeks old (Janvier Labs). Animals were acclimatised to the vivarium for 1–2 weeks prior to use in experiments. SCID mice were infected with 10^4 trypomastigotes and all other mice were infected with 10^3 trypomastigotes. Hamsters were infected with 5 x 10^3 or 10^5 trypomastigotes. Inocula were given as intra-peritoneal (i.p.) injections of 0.2 mL in sterile 1X Dulbecco's Phosphate Buffered Saline (DPBS).

Animals were maintained under specific pathogen-free conditions in individually ventilated cages, with 2–3 hamsters and 5–6 mice per cage, with a 12 hour light/dark cycle and *ad libitum* food and water. An investigator who was blinded to the infection/control group allocations assigned individual animals to cages. No randomisation procedure was used to allocate animals to infection or control groups. Animals were checked visually twice daily and weighed on average every two weeks. Humane end-points were any of the following: loss of >20% body weight, unwillingness to move, reluctance to feed and drink freely for >6 hours, loss of balance. No animals in this study reached any of the humane end-points.

Bioluminescence imaging

Hamsters were injected with 150 mg/kg d-luciferin i.p., then anaesthetized using 2.5% (v/v) gaseous isoflurane in oxygen. To measure bioluminescence, hamsters were placed in a Lumina II In Vivo Imaging System (IVIS) (PerkinElmer) and images were acquired 10–20 minutes after dluciferin administration using LivingImage v4.7 (PerkinElmer). A custom 3D printed nosecone was used for continuous isoflurane delivery during imaging sessions (S1 Data). Exposure times varied between 30 seconds and 5 minutes, depending on bioluminescence signal intensity. After imaging, animals were revived and returned to cages. For *ex vivo* imaging, hamsters were injected with 150 mg/kg d-luciferin i.p., then sacrificed by ex-sanguination under terminal anaesthesia (400–600 mg/kg Pentobarbital Sodium) 7 minutes later. Trans-cardiac perfusion was performed with 20 mL 0.3 mg/mL d-luciferin in PBS. Organs and tissues were transferred to culture dishes, soaked in 0.3 mg/mL d-luciferin in PBS, and then imaged using the IVIS Lumina II. One hamster from the low inoculum group was excluded from further analysis because a bioluminescence signal was never detected, indicating that infection did not become established.

To estimate total parasite loads in live hamsters, regions of interest (ROIs) were drawn using Living Image v4.7 to quantify bioluminescence expressed as total flux (photons/second [p/s]). The detection threshold for *in vivo* imaging was determined using uninfected control hamsters [34].

Ex vivo images of tissues and organs were scored for the presence of *T. cruzi* using a detection threshold for infection foci of at least 10 contiguous bioluminescent pixels of radiance $\geq 3x10^3 \text{ p/s/cm}^2/\text{sr}$ [19]. Ex vivo parasite loads were inferred from tissue/organ-specific bioluminescence total flux (p/s) after subtracting the mean + 2SDs of matching samples from

uninfected control hamsters (n = 7) to account for differences in background luminescence signal amongst different tissue and organ types. Tissue/organ-specific infection intensities were calculated as the fold change in bioluminescence radiance ($p/s/cm^2/sr$), which normalises for differences in tissue sample surface areas, compared with the mean radiance for the matching tissue/organ type measured for uninfected control hamsters (n = 7).

Histopathology

Tissue samples were fixed in Glyo-Fixx (Epredia) for 24–72 hours, then dehydrated in an ethanol series, cleared in xylene (Sigma), and embedded in paraffin for histomorphometric analysis [9,19]. Five-micron tissue sections were stained with Haematoxylin and Eosin (H&E), picrosirius red or Masson's trichrome and analysed using a DFC295 camera attached to a DM3000 light-emitting diode microscope (Leica) as described [9,19,35,36]. Images were acquired of 10–15 randomly selected microscope fields of view (FOV, area at 400X magnification = 2.66 x $10^5 \mu m^2$) per animal. Images were digitised using Leica Application Suite v4.5.0. For the analysis of tissue inflammatory infiltrates, the total number of nuclei were counted in H&E-stained sections. Myositis (skeletal muscle) and myocarditis (heart) scores were calculated as the average cellularity (mean number of nuclei per FOV) for each animal. A significant increase in cellularity compared to uninfected controls was considered indicative of inflammation. For the fibrosis index, the percentage of total FOV area positive for collagen in picrosirius red- or Masson's trichrome stained sections (red or blue pixels respectively) was quantified, also in Leica Application Suite v4.5.0. A significant increase in collagen content compared to uninfected controls was considered indicative of inflammation.

Limb adduction evaluation

Images from the *in vivo* imaging sessions were adjusted in Living Image v4.7 (PerkinElmer) to remove the bioluminescence overlay. The line measurement tool was then used to measure the distance between the centres of the two forepaws and between the centres of the two hindpaws.

Statistics

Individual animals were used as the unit of analysis. Groups were compared using Student's *t*-test or one-way ANOVA, with Tukey's post-hoc correction in GraphPad Prism v.8.

Results

In vivo bioluminescence imaging permits long-term tracking of *T. cruzi* infection in hamsters

We infected hamsters with either a low (5,000) or high (100,000) inoculum of *T. cruzi* TcVI-CLBR trypomastigotes expressing the red-shifted luciferase reporter *PPy*RE9h [34]. The resulting infections were monitored regularly by *in vivo* bioluminescence imaging (Fig 1A and 1B). For both inocula, parasite burdens proceeded through a typical acute phase wave that transitioned into a stable, long-term chronic infection. In the high inoculum group, the acute phase was 3-fold more intense and its peak occurred at 2 weeks post-infection (p.i.) rather than 4 weeks p.i. for the low inoculum group (Fig 1B). Spatially, *in vivo* imaging indicated that the acute phase infection was highly concentrated near to the injection site in the abdomen, but from 8 weeks p.i. onwards, the bioluminescence signal became widely distributed, multifocal and dynamic between analysis time-points (Fig 1A). The size of the inoculum did not affect the level at which chronic parasite burdens stabilised, nor the spatio-temporal



Fig 1. Serial evaluation of *T. cruzi* **CLBR infection in Syrian hamsters by** *in vivo* **bioluminescence imaging. A, B:** Course of transgenic bioluminescent TcVI-CLBR infection in female Syrian hamsters tracked by *in vivo* imaging. **A:** Panels show an example series of ventral images for one hamster per group taken at the indicated times post-infection. Log-scale pseudocolour heat-map shows intensity of bioluminescence as a proxy for parasite load; minimum and maximum radiances are indicated. **B:** Chart shows the changes in mean \pm S.E.M., total bioluminescence over time for hamsters inoculated with 5 x 10³ (low inoculum, n = 3) or 1 x 10⁵ (high inoculum, n = 8) parasites. Threshold lines are the mean (dashes) and the mean +2SD (dots) of background signal obtained for uninfected control hamsters (n = 7). **C:** Evaluation of organ sizes/weights and the number of faecal pellets present in the colon at 152–182 days post-infection (high inoculum group (n = 8–9 with the exception of GI mesentery, n = 5, and faecal pellets n = 6), uninfected control group (n = 8, with the exception of GI mesentery and faecal pellets, n = 4). Asterisks indicate *p*-values for comparisons with the uninfected control group in multiple unpaired *t*-tests (*** *p*<0.001, * *p*<0.05).

https://doi.org/10.1371/journal.pntd.0012278.g001

distribution of the infection. For further analyses, we focussed on the high inoculum approach because acute infections in the low inoculum group tended to be more variable.

Monitoring of animals identified gait abnormalities gradually developing during the chronic phase of infection, with pronounced hindlimb rigidity. Muscle mass was observed to

be within normal limits at post-mortem. With respect to potential features of human Chagas disease, we observed significant splenomegaly, but no sign of cardiomegaly, digestive mega-syndromes or constipation at the experimental end-point of 152–182 days p.i. (Fig 1C).

Skin is the primary reservoir of chronic parasite persistence in hamsters

T. cruzi displays a very broad cell and tissue tropism, but *in vivo* imaging does not permit unambiguous localisation of signals to organs. Therefore, to better understand infection dynamics in this model we conducted *ex vivo* bioluminescence imaging post-mortem, at the acute-chronic transition (34–38 days p.i.) and in the established chronic phase (152–182 days p.i.) (Fig 2A). At 34–38 days p.i., infection was almost universally detected across all the tissues and organs that were sampled (Fig 2A, 2B, 2D and 2F). In terms of absolute parasite loads (total bioluminescent flux/sec), the skin harboured the greatest proportion of parasites (58.0% ±19.2%) followed by several fat-rich organs each containing 5–10% of the total (GI mesentery, genito-urinary system, subcutaneous adipose, visceral adipose) (Fig 2D). Infection intensity, which takes into account the large variation in organ size, varied over more than three orders of magnitude. Of note, parasite loads measured in this way were relatively high in the skin, adipose, GI mesentery and genito-urinary system, moderately high in striated muscles (heart, skeletal muscle, abdomen/peritoneum) and relatively low in the GI tract, liver and spleen (Figs 2F and S1).

In the established chronic phase, infection became undetectable in the majority of sites in the majority of individual animals (Fig 2A, 2C, 2E and 2G). It should be noted that the limit of detection is <20 parasites in the mouse colon [37], but the larger size of hamster organs likely reduces detection sensitivity. Persistent infection was mainly restricted to the skin and subcutaneous adipose tissue, with sporadic, usually small foci of infection in a few other organs, most often the skeletal muscle, spleen and lungs. In contrast to mouse models [9,15], parasites were only rarely detected in the GI tract during the chronic stage. Considering absolute numbers, the skin harboured a large majority of the total parasites (93.4% \pm 3.8%) (Fig 2D and 2E) and also had the highest infection intensity (Fig 2G). Skin parasites could be detected in any skin region, and in the chronic phase they were typically present in approximately 20 to 30 discrete foci of infection (Figs 2A and S1).

Occurrence of a hindlimb spastic diplegia-like syndrome in the hamster infection model

Cardiomyopathy is the most serious adverse clinical consequence of chronic *T. cruzi* infection in humans. Histopathology analysis of heart tissue samples from hamsters at 5 months p.i. indicated that inflammation and collagen content (a measure of fibrosis) were not significantly increased compared to uninfected control animals at the group mean level (Fig 3). However, some animals did display clear evidence of oedema, focal and diffuse inflammatory foci, and intermuscular fibro-fatty tissue replacement.

To better understand the gait abnormality that arose in the hamsters, we assessed its development over time (Fig 4A). This identified excessive bilateral hindlimb adduction (movement towards the body's midline), which was clearly progressive over the course of chronic infection and resembled the "scissor gait" seen in a number of human neuromuscular disorders, such as cerebral palsy [38,39]. The phenotype was highly significant and developed in 100% of chronically infected hamsters (Fig 4D). Histopathological analysis of hindlimb skeletal muscle identified significant perivascular and muscle parenchymal inflammation, fibrosis and fatty degeneration (Fig 5). To investigate whether the gait abnormality might be linked to the myositis (muscle inflammation) and intense fibrosis, we investigated if these histopathological



Fig 2. Organ-specific infection profiles of *T. cruzi* **CLBR in Syrian hamsters.** Quantification of organ/tissue-specific parasite loads and densities in TcVI-CLBR-infected Syrian hamsters using *ex vivo* bioluminescence imaging at 34–38 days (n = 6) and 152–182 days (n = 9) post-infection. Images (**A**) show bioluminescence intensities for one representative animal per group. Layout of samples shown on the uninfected control image: 1 = liver (partial), 2 = lungs, 3 = heart (bisected), 4 = kidneys, 5 = spleen, 6 = visceral adipose, 7 = female reproductive system, 8 = bladder, 9 = skeletal muscle (from hindleg), 10 = oesophagus, 11 = stomach, 12 = small intestine, 13 = gastro-intestinal mesentery, 14 = peritoneum/abdominal muscle, 15 = caecum, 16 = colon, 17 = subcutaneous adipose, 18 = visceral adipose, 19 = skin. **B,C**: Quantification of total bioluminescent flux (photons/sec) as a proxy of parasite loads for each organ and tissue type at 34–38 days (**B**) and 152–182 days (**C**) post-infection. **D,E**: Charts show the average proportional distributions of total measured *ex vivo*

PLOS Neglected Tropical Diseases | https://doi.org/10.1371/journal.pntd.0012278 June 21, 2024

7/17

parasite loads (total flux, p/s) across the analysed set of organ and tissue types at 34-38 days (**D**) and 152-182 days (**E**) post-infection. **F,G**: Quantification of bioluminescent radiance (photons/sec/cm²/sr) as a proxy of infection intensity for each organ and tissue type expressed as fold change vs. matching organs or tissues from uninfected controls at 34-38 days (**F**) and 152-182 days (**G**) post-infection. Dashed line indicates detection threshold. In **B**, **C**, **F** and **G**, bars show the mean + S.E.M., and grey circles are data for individual hamsters.

https://doi.org/10.1371/journal.pntd.0012278.g002

features developed in a range of murine *T. cruzi* infection models (parasite strains TcI-JR and TcVI-CLBR in BALB/c, C57BL/6 and C3H/HeN mice), none of which develop the diplegia symptoms seen in hamsters. Significant myositis was present in 5 of the 6 parasite-mouse strain combinations, TcI-JR in C3H/HeN mice being the exception, and significant fibrosis was present in all 6 models (Fig 6). Both phenotypes occurred with a range of severities, with the most affected mouse models being broadly comparable to the hamster model: cellularity (myositis) increased 3-fold in TcVI-CLBR:C3H/HeN vs 4-fold for TcVI-CLBR:Hamster, and collagen content (fibrosis) increased ~4.5 fold in TcVI-CLBR:C3H/HeN vs 2-fold for TcVI-CLBR:Hamster. Fatty degeneration and parasite nests were also occasionally observed in murine skeletal muscle tissue. Thus, the diplegic gait in hamsters is highly unlikely to result



Fig 3. Histopathological analysis of cardiac muscle tissue from *T. cruzi*-infected Syrian hamsters. Quantification of cellularity (A) and collagen content (B) at 152–182 days post-infection as markers of chronic myocardial inflammation and fibrosis, respectively. Bars show the means + S.E.M., and grey circles are data for individual hamsters. Uninfected control (n = 7), infected (n = 9). Images show representative example sections stained with haematoxylin and eosin to reveal and inflammatory infiltrates (black arrows) (A), and sections stained with picro-sirius red to reveal excess collagen deposition (white arrows) and adipose replacement (asterisks) (B). Magnification 400X, scale bar = 50 μ m.

https://doi.org/10.1371/journal.pntd.0012278.g003



Fig 4. Evaluation of limb adduction in *T. cruzi-***infected Syrian hamsters. A:** Example ventral view images of an individual hamster under transient anaesthesia (gaseous isoflurane) at the indicated day post-infection, with an age-matched uninfected control. Red marker lines highlight the inter-hindpaw distances. **B–E**: Quantification of the distance between forepaws (**B**, **C**) and hindpaws (**D**, **E**). In **B** and **D**, data are shown as the mean \pm S.E.M. over time, uninfected control (n = 2), infected (n = 4). In **C** and **E**, data are the mean \pm S.E.M at the endpoint, with grey circles showing the measurements for individual animals, uninfected control (n = 5), infected (n = 7) (**C**, **E**).

https://doi.org/10.1371/journal.pntd.0012278.g004

directly from chronic parasite persistence, skeletal muscle inflammation or fibrosis, because these are all also common features of murine infections.

Discussion

Bioluminescence imaging has been widely applied to *T. cruzi* murine infections, but its utility to monitor Chagas disease progression in larger rodents has not been reported. Our aims were to investigate if this technology could provide new insights into parasite distribution patterns during long-term *T. cruzi* infection of hamsters and to establish if there was a link with pathological outcomes. The results demonstrate that *in vivo* imaging, using the highly sensitive system described here, is able to facilitate longitudinal monitoring of chronic *T. cruzi* infections in hamsters. Furthermore, *ex vivo* imaging revealed organ and tissue parasite tropism during both acute and chronic stage infections, and identified the reservoir sites within the hamster that provide a niche for long-term parasite survival.



Fig 5. Histopathological analysis of quadriceps tissue from *T. cruzi*-infected Syrian hamsters. Quantification of cellularity (A) and collagen content (B) at 152–182 days post-infection as markers of chronic skeletal muscle inflammation and fibrosis, respectively. Bars show the means + S.E.M., and grey circles are data for individual hamsters. Uninfected control (n = 7), infected (n = 9). Images show representative example sections stained with haematoxylin and eosin to reveal and inflammatory infiltrates (black arrows) (A), and sections stained with picro-sirius red to reveal excess collagen deposition (white arrows) and adipose replacement (asterisks) (B). Magnification 400X, scale bar = 50 μ m. *P*-values shown are from Student's *t*-tests.

https://doi.org/10.1371/journal.pntd.0012278.g005

There were both similarities and differences in comparison with typical infection profiles in mice, with one commonality being the pan-tropism of acute stage infections [34,40–44]. In terms of the heart, hamsters did not have detectable chronic infections localised to this organ and signs of cardiac pathology were only observed in a small subset of animals. Cardiac pathology typically is present in mice, although severity can vary substantially between parasite and mouse strains as, well as between individual animals, even when they are inbred [8–10,45]. The observation of a low rate of cardiomyopathy in hamsters may be significant, given that they are an outbred model, and only a minority of humans infected with *T. cruzi* develop heart disease symptoms [46]. Previous studies in hamsters have reported more consistent cardiomyopathy as well as mortality rates ranging from 12–80% [27–29,31–33,47]. However, here we observed 100% survival. This could reflect differences in virulence between strains because TcVI-CLBR also has low mortality, <10%, in mice in our laboratory [9]. In addition, the 5–6 month post-infection period to which our study was limited may have been too short for cardiac pathology to develop more reproducibly.



Fig 6. Histopathological analysis of skeletal muscle tissue from multiple murine *T. cruzi* models. A: Representative hindlimb muscle sections stained with haematoxylin and eosin, magnification 400X, scale bar = 100 μ m. B: Quantitative histopathological analysis of hindlimb muscle samples obtained at 154–174 days post-infection from the following groups: TcVI-CLBR-BALB/c (n = 9), TcVI-CLBR-C57BL/6 (n = 10), TcVI-CLBR-C3H/HeN (n = 10), TcI-JR-BALB/c (n = 10), TcI-JR-C57BL/6 (n = 10), TcI-JR-C57BL/6 (n = 10), TcI-JR-C57BL/6 (n = 10), TcI-JR-C3H/HeN (n = 8), uninfected control BALB/c, C57BL.6 and C3H/HeN (all n = 10). Myositis score is the number of nuclei per image (6 x 10⁴ μ m²). In **A** and **B**, black and white arrows indicate examples of parasite nests and areas of adipose replacement respectively. **C**: Representative hindlimb muscle sections stained with Masson's trichrome, magnification 400X, scale bar = 100 μ m. **D**: Quantification of collagen content (% blue area) as a marker of skeletal muscle fibrosis severity in same groups as in **B**. Data are the means +S.E.M. and are from two independent experiments. Data for the equivalent analysis of tissue pathology in hamsters infected with TcVI-CLBR is shown for comparison. Asterisks indicate *p*-values for one way ANOVA comparisons between infection status groups, with each mouse strain analysed independently (* *P* < 0.05; ** *P* < 0.01; *** *P* < 0.001).

https://doi.org/10.1371/journal.pntd.0012278.g006

Regarding the digestive tract, localised chronic infections were absent, apart from a few very weak bioluminescence signals in one or two animals. No signs of organ enlargement or constipation, characteristic of digestive Chagas disease, were found. This contrasts with murine infections, where the GI tract is a major and almost universal site of long-term parasite persistence and chronic transit delays linked to the colon are significant in some parasitemouse strain combinations [9,48–51]. GI tract samples from digestive Chagas disease patients have high frequencies of parasite DNA/antigen positivity [52–55]. Overall, these data indicate that Syrian hamster infection with TcVI-CLBR is not a suitable model to study the cardiac or the digestive form of Chagas disease.

Two findings from the study are particularly noteworthy. Firstly, the predominance of the skin as a chronic reservoir of infection, and secondly, the hindlimb diplegic gait phenotype. In mouse models, we have shown that chronic skin infection is detectable in 80-90% of mice across multiple combinations of T. cruzi and host genetic strains [19,37]. Whereas in mice the skin and GI tract have a similar reservoir status, in the hamster our data indicate that the skin and subcutaneous adipose tissue harbour almost the entire parasite burden in the animal. Acute cutaneous manifestations of Chagas disease reported in humans include localised inoculation 'chagomas'-swellings associated with initial parasite entry sites [2] and, very rarely, more disseminated lesions [56]. Skin lesions may also occur in cases of 're-activation' of a chronic infection in which immunosuppression, for example caused by HIV co-infection or post-transplant medications, causes expansion and spread of the parasite burden [57,58]. Although chronic T. cruzi infection in immunocompetent people is not associated with cutaneous pathology, this does not rule out the existence of a clinically silent, skin-resident parasite population. This would have an obvious evolutionary advantage for T. cruzi in terms of ensuring onward transmission to its blood-feeding insect vector and would be in keeping with the biology of related trypanosomatids, such as T. brucei and Leishmania donovani [59,60]. It may also be relevant that the skin is the most common site affected by adverse reactions to the antiparasitic drugs that are used to treat Chagas disease [61]. As well as establishing how frequent chronic skin infections are in humans, it will be important to investigate which cell types in the skin are parasitised and whether transmission to triatomines involves uptake of parasites from the tissue parenchyma or only the vasculature.

Skeletal muscle is a preferred site of infection for numerous human parasites e.g. Sarcocystis, Toxoplasma, Brugia, Ancylostoma, and Trichinella spp., [62]. Parasitism of this tissue is common in T. cruzi infected mice, though the frequency and intensity appears to be rather model-dependent [9,37,49,63]. We observed hindlimb skeletal muscle-localised parasite bioluminescence in all hamsters at 34-38 days p.i., and in five of nine (55.6%) hamsters at 152-182 days p.i., although the infection intensity was relatively low compared to other sites. This tissue was characterised by significant chronic parenchymal and perivascular inflammation and fibrosis, which was also present in all six of the mouse models that we analysed. The hindlimb hypertonia and diplegic gait were hamster-specific; no locomotion defects were observed in mice, implying that tissue parasitism, inflammation and fibrosis are insufficient explanations. A motor neuron-related aetiology is probably the most suitable hypothesis for future work. Others have reported limb paralysis in *T. cruzi*-infected mice [64,65], but that is clearly a different phenomenon from the one we have observed [66]. While neither type of locomotive dysfunction have been linked to T. cruzi infection in humans, there is evidence of peripheral nervous system involvement in some Chagas disease cohorts at frequencies of 10% [67] to 27% [68]. Skeletal muscle denervation, neuro-muscular junction, microvascular, mitochondrial and metabolic abnormalities have all been described [69-73], so it would be interesting to analyse whether these occur in T. cruzi-infected hamsters. Spastic diplegia in cerebral palsy is caused by congenital CNS lesions that impact on motor function [66] and T. cruzi infection

can extend to the CNS in severe cases, usually in association with immunosuppression [21,74]. A recent systematic review of literature encompassing 138 Chagas disease patients with CNS involvement found that 70% presented with a motor deficit, but only one case study specifically reporting spasticity [21].

In conclusion, further investigation of the mechanism underlying the gait defect we observe in our hamster model should consider both peripheral and central nervous system involvement. Neuropathological aspects of Chagas disease are evidently under-studied and the potential translational value of the hamster model developed in this study should be explored further. Lastly, this infection model may also prove relevant to wider research on spastic diplegia (bilateral lower limb muscle tightness), a common presentation of cerebral palsy that lacks a predictive small animal model [66].

Supporting information

S1 Fig. *Ex vivo* bioluminescence images for hindlimb skeletal muscle and skin samples from chronic *T. cruzi*-infected hamsters. Images for each hamster were taken immediately post-mortem at at 152–182 days post-infection. Numbers are individual animal codes. Images from one representative not infected control are shown for comparison. Log-scale pseudocolour heat-map shows intensity of bioluminescence as a proxy for parasite load; minimum and maximum radiances are indicated. Red arrows highlight small foci of bioluminescence. (TIF)

S1 Data. Source files for 3D printed hamster anaesthesia delivery nosecone. (ZIP)

S2 Data. Source data for figures in the manuscript. (XLSX)

Acknowledgments

The authors thank the LSHTM Biological Services Facility and Veterinary staff for technical support and helpful discussions.

Author Contributions

Conceptualization: Martin C. Taylor, John M. Kelly, Michael D. Lewis.

Data curation: Harry Langston, Amanda Fortes Francisco, Michael D. Lewis.

Formal analysis: Harry Langston, Amanda Fortes Francisco, Michael D. Lewis.

Funding acquisition: John M. Kelly, Michael D. Lewis.

Investigation: Harry Langston, Amanda Fortes Francisco, Ciaran Doidge, Archie A. Khan, Shiromani Jayawardhana, Michael D. Lewis.

Methodology: Harry Langston, Amanda Fortes Francisco, Michael D. Lewis.

Project administration: Martin C. Taylor, John M. Kelly, Michael D. Lewis.

Resources: Chrissy H. Roberts, Michael D. Lewis.

Supervision: John M. Kelly, Michael D. Lewis.

Validation: Harry Langston, Amanda Fortes Francisco, Ciaran Doidge.

Visualization: Harry Langston, Ciaran Doidge.

Writing - original draft: Michael D. Lewis.

Writing – review & editing: Harry Langston, Amanda Fortes Francisco, Archie A. Khan, Martin C. Taylor, John M. Kelly, Michael D. Lewis.

References

- Gómez-Ochoa SA, Rojas LZ, Echeverría LE, Muka T, Franco OH. Global, regional, and national trends of Chagas disease from 1990 to 2019: comprehensive analysis of the global burden of disease study. Global Heart. 2022; 17(1). https://doi.org/10.5334/gh.1150 PMID: 36051318
- Pérez-Molina JA, Molina I. Chagas disease. Lancet (London, England). 2018; 391(10115):82–94. https://doi.org/10.1016/S0140-6736(17)31612-4 PMID: 28673423
- Marin-Neto JA, Cunha-Neto E, Maciel BC, Simoes MV. Pathogenesis of chronic Chagas heart disease. Circulation. 2007; 115(9):1109–23. <u>https://doi.org/10.1161/CIRCULATIONAHA.106.624296</u> PMID: 17339569
- Bonney KM, Luthringer DJ, Kim SA, Garg NJ, Engman DM. Pathology and Pathogenesis of Chagas Heart Disease. Annual Review of Pathology: Mechanisms of Disease. 2019; 14(1):421–47. <u>https://doi.org/10.1146/annurev-pathol-020117-043711</u> PMID: 30355152
- Machado FS, Dutra WO, Esper L, Gollob KJ, Teixeira MM, Factor SM, et al. Current understanding of immunity to *Trypanosoma cruzi* infection and pathogenesis of Chagas disease. Seminars in Immunopathology. 2012; 34(6):753–70.
- Jansen AM, Xavier S, Roque ALR. *Trypanosoma cruzi* transmission in the wild and its most important reservoir hosts in Brazil. Parasit Vectors. 2018; 11(1):502. <u>https://doi.org/10.1186/s13071-018-3067-2</u> PMID: 30189896
- Chatelain E, Konar N. Translational challenges of animal models in Chagas disease drug development: a review. Drug Design, Development and Therapy. 2015; 9:4807. <u>https://doi.org/10.2147/DDDT.</u> S90208 PMID: 26316715
- Andrade SG. Influence of *Trypanosoma cruzi* strain on the pathogenesis of chronic myocardiopathy in mice. Memorias do Instituto Oswaldo Cruz. 1990; 85(1):17–27. <u>https://doi.org/10.1590/s0074-</u>02761990000100003 PMID: 2215229
- **9.** Lewis MD, Francisco AF, Taylor MC, Jayawardhana S, Kelly JM. Host and parasite genetics shape a link between *Trypanosoma cruzi* infection dynamics and chronic cardiomyopathy. Cell Microbiol. 2016; 18(10):1429–43.
- Postan M, Bailey JJ, Dvorak JA, McDaniel JP, Pottala EW. Studies of *Trypanosoma cruzi* Clones in Inbred Mice: III. Histopathological and Electrocardiographical Responses to Chronic Infection. The American Journal of Tropical Medicine and Hygiene. 1987; 37(3):541–9.
- 11. Rodriguez HO, Guerrero NA, Fortes A, Santi-Rocca J, Gironès N, Fresno M. *Trypanosoma cruzi* strains cause different myocarditis patterns in infected mice. Acta Tropica. 2014; 139:57–66.
- Schlemper BR Jr, Avila CM, Coura JR, Brener Z. Course of infection and histopathological lesions in mice infected with seventeen *Trypanosoma cruzi* strains isolated from chronic patients. Rev Soc Bras Med Trop. 1983; 16:23–30.
- Soares MB, de Lima RS, Rocha LL, Vasconcelos JF, Rogatto SR, dos Santos RR, et al. Gene expression changes associated with myocarditis and fibrosis in hearts of mice with chronic chagasic cardiomyopathy. J Infect Dis. 2010; 202(3):416–26. https://doi.org/10.1086/653481 PMID: 20565256
- Prado CM, Celes MR, Malvestio LM, Campos EC, Silva JS, Jelicks LA, et al. Early dystrophin disruption in the pathogenesis of experimental chronic Chagas cardiomyopathy. Microbes and Infection. 2012; 14 (1):59–68. https://doi.org/10.1016/j.micinf.2011.08.010 PMID: 21914488
- Khan AA, Langston HC, Costa FC, Olmo F, Taylor MC, McCann CJ, et al. Local association of *Trypano-soma cruzi* chronic infection foci and enteric neuropathic lesions at the tissue micro-domain scale. PLoS Pathog. 2021; 17(8):e1009864.
- do Carmo Neto JR, Vinicius da Silva M, Braga YLL, Florencio da Costa AW, Fonseca SG, Nagib PRA, et al. Correlation between intestinal BMP2, IFNy, and neural death in experimental infection with *Trypanosoma cruzi*. PLOS ONE. 2021; 16(2):e0246692.
- Campos CF, Cangussú SD, Duz ALC, Cartelle CT, Noviello MdL, Veloso VM, et al. Enteric Neuronal Damage, Intramuscular Denervation and Smooth Muscle Phenotype Changes as Mechanisms of Chagasic Megacolon: Evidence from a Long-Term Murine Model of *Trypanosoma cruzi* Infection. PLoS ONE. 2016; 11(4):e0153038.

- Oda JY, Belém MO, Carlos TM, Gouveia R, Luchetti BFC, Moreira NM, et al. Myenteric neuroprotective role of aspirin in acute and chronic experimental infections with *Trypanosoma cruzi*. Neurogastroenterology and Motility. 2017; 29(10):1–13.
- Lewis MD, Francisco AF, Jayawardhana S, Langston H, Taylor MC, Kelly JM. Imaging the development of chronic Chagas disease after oral transmission. Scientific Reports. 2018; 8(1):11292. <u>https://doi.org/ 10.1038/s41598-018-29564-7 PMID</u>: 30050153
- Berkowitz AL, Raibagkar P, Pritt BS, Mateen FJ. Neurologic manifestations of the neglected tropical diseases. J Neurol Sci. 2015; 349(1–2):20–32. https://doi.org/10.1016/j.jns.2015.01.001 PMID: 25623803
- Shelton WJ, Gonzalez JM. Outcomes of patients in Chagas disease of the central nervous system: A systematic review. Parasitology. 2024:1–28. https://doi.org/10.1017/S0031182023001117 PMID: 37987164
- Nieto A, Domínguez-Bernal G, Orden JA, De La Fuente R, Madrid-Elena N, Carrión J. Mechanisms of resistance and susceptibility to experimental visceral leishmaniosis: BALB/c mouse versus syrian hamster model. Veterinary Research. 2011; 42(1):39. <u>https://doi.org/10.1186/1297-9716-42-39</u> PMID: 21345200
- Toth K, Lee SR, Ying B, Spencer JF, Tollefson AE, Sagartz JE, et al. STAT2 knockout Syrian hamsters support enhanced replication and pathogenicity of human adenovirus, revealing an important role of type I interferon response in viral control. PLoS Pathogens. 2015; 11(8):e1005084.
- Hutton ML, Mackin KE, Chakravorty A, Lyras D. Small animal models for the study of *Clostridium difficile* disease pathogenesis. FEMS Microbiology Letters. 2014; 352(2):140–9.
- Lee C-Y, Lowen AC. Animal models for SARS-CoV-2. Current Opinion in Virology. 2021; 48:73–81. https://doi.org/10.1016/j.coviro.2021.03.009 PMID: 33906125
- Miao J, Chard LS, Wang Z, Wang Y. Syrian Hamster as an Animal Model for the Study on Infectious Diseases. Front Immunol. 2019; 10:2329.
- Cabrine-Santos M, Lages Silva E, Chapadeiro E, Ramírez LE. *Trypanosoma cruzi*: characterization of reinfection and search for tissue tropism in hamsters (*Mesocricetus auratus*). Experimental Parasitology. 2001; 99(3):160–7.
- Ramírez LE, Lages-Silva E, Soares Júnior JM, Chapadeiro E. The hamster (*Mesocricetus auratus*) as experimental model in Chagas' disease: parasitological and histopathological studies in acute and chronic phases of *Trypanosoma cruzi* infection. Rev Soc Bras Med Trop. 1994; 27(3):163–9.
- Ramires FJ, Salemi VM, Ianni BM, Fernandes F, Martins DG, Billate A, et al. Aldosterone antagonism in an inflammatory state: evidence for myocardial protection. J Renin Angiotensin Aldosterone Syst. 2006; 7(3):162–7. https://doi.org/10.3317/jraas.2006.026 PMID: 17094053
- Bilate AM, Teixeira PC, Ribeiro SP, Brito T, Silva AM, Russo M, et al. Distinct outcomes of *Trypanosoma cruzi* infection in hamsters are related to myocardial parasitism, cytokine/chemokine gene expression, and protein expression profile. J Infect Dis. 2008; 198(4):614–23.
- Bilate AM, Salemi VM, Ramires FJ, de Brito T, Silva AM, Umezawa ES, et al. The Syrian hamster as a model for the dilated cardiomyopathy of Chagas' disease: a quantitative echocardiographical and histopathological analysis. Microbes and Infection. 2003; 5(12):1116–24. https://doi.org/10.1016/j.micinf. 2003.07.001 PMID: 14554253
- Chapadeiro E, Silva EL, Silva AC, Fernandes P, Ramirez LE. Cardiac neuronal depopulation in hamsters (*Mesocricetus auratus*) chronically infected with *Trypanosoma cruzi*. Rev Soc Bras Med Trop. 1999; 32(1):35–9.
- Pimentel WS, Ramires FJ, Lanni BM, Salemi VM, Bilate AM, Cunha-Neto E, et al. The effect of betablockade on myocardial remodelling in Chagas' cardiomyopathy. Clinics (Sao Paulo). 2012; 67 (9):1063–9. https://doi.org/10.6061/clinics/2012(09)14 PMID: 23018305
- 34. Lewis MD, Francisco AF, Taylor MC, Burrell-Saward H, McLatchie AP, Miles MA, Kelly JM. Bioluminescence imaging of chronic *Trypanosoma cruzi* infections reveals tissue-specific parasite dynamics and heart disease in the absence of locally persistent infection. Cell Microbiol. 2014; 16(9):1285–300.
- 35. Caliari MV, Machado RdP, de Lana M, Cajá RAF, Carneiro CM, Bahia MT, et al. Quantitative analysis of cardiac lesions in chronic canine chagasic cardiomyopathy. Rev I Med Trop. 2002; 44:273–8. https://doi.org/10.1590/s0036-46652002000500008 PMID: 12436168
- Caliari MV. Princípios de Morfometria Digital: KS300 para iniciantes. Belo Horizonte, Brazil: Editora UFMG; 1997.
- Ward AI, Lewis MD, Khan AA, McCann CJ, Francisco AF, Jayawardhana S, et al. *In Vivo* Analysis of *Trypanosoma cruzi* Persistence Foci at Single-Cell Resolution. mBio. 2020; 11(4):e01242–20.
- 38. Koman LA, Smith BP, Shilt JS. Cerebral palsy. The Lancet. 2004; 363(9421):1619–31.
- **39.** Kheder A, Nair KPS. Spasticity: pathophysiology, evaluation and management. Practical Neurology. 2012; 12(5):289–98. https://doi.org/10.1136/practneurol-2011-000155 PMID: 22976059

- Hyland KV, Asfaw SH, Olson CL, Daniels MD, Engman DM. Bioluminescent imaging of *Trypanosoma cruzi* infection. International Journal for Parasitology. 2008; 38(12):1391–400.
- Lenzi HL, Oliveira DN, Lima MT, Gattass CR. *Trypanosoma cruzi*: Paninfectivity of CL Strain during Murine Acute Infection. Experimental Parasitology. 1996; 84(1):16–27.
- 42. Guarner J, Bartlett J, Zaki SR, Colley DG, Grijalva MJ, Powell MR. Mouse model for Chagas disease: immunohistochemical distribution of different stages of *Trypanosoma cruzi* in tissues throughout infection. The American Journal of Tropical Medicine and Hygiene. 2001; 65(2):152–8.
- 43. Postan M, Dvorak JA, McDaniel JP. Studies of *Trypanosoma cruzi* Clones in Inbred Mice: I. A Comparison of the Course of Infection of C3H/HEN- Mice with Two Clones Isolated from a Common Source. The American Journal of Tropical Medicine and Hygiene. 1983; 32(3):497–506.
- Melo R, Brener Z. Tissue tropism of different *Trypanosoma cruzi* strains. J Parasitol. 1978; 64(3):475– 82.
- Francisco AF, Jayawardhana S, Taylor MC, Lewis MD, Kelly JM. Assessing the Effectiveness of Curative Benznidazole Treatment in Preventing Chronic Cardiac Pathology in Experimental Models of Chagas Disease. Antimicrobial Agents and Chemotherapy. 2018; 62(10). https://doi.org/10.1128/AAC. 00832-18 PMID: 30082291
- 46. Chadalawada S, Sillau S, Archuleta S, Mundo W, Bandali M, Parra-Henao G, et al. Risk of Chronic Cardiomyopathy Among Patients With the Acute Phase or Indeterminate Form of Chagas Disease: A Systematic Review and Meta-analysis. JAMA Network Open. 2020; 3(8):e2015072–e. https://doi.org/10. 1001/jamanetworkopen.2020.15072 PMID: 32865573
- 47. dos Santos VM, de Lima MA, Cabrine-Santos M, de Stefani Marquez D, de Araújo Pereira G, Lages-Silva E, Ramírez LE. Functional and histopathological study of the pancreas in hamsters (*Mesocricetus auratus*) infected and reinfected with *Trypanosoma cruzi*. Parasitol Res. 2004; 94(2):125–33.
- Khan AA, Langston HC, Walsh L, Roscoe R, Jayawardhana S, Francisco AF, et al. Enteric nervous system regeneration and functional cure of experimental digestive Chagas disease with trypanocidal chemotherapy. Nat Commun. 2024 May 23; 15(1):4400. https://doi:10.1038/s41467-024-48749-5
- Santi-Rocca J, Fernandez-Cortes F, Chillón-Marinas C, González-Rubio M-L, Martin D, Gironès N, Fresno M. A multi-parametric analysis of *Trypanosoma cruzi* infection: common pathophysiologic patterns beyond extreme heterogeneity of host responses. Scientific Reports. 2017; 7(1):1–12.
- Silberstein E, Serna C, Fragoso SP, Nagarkatti R, Debrabant A. A novel nanoluciferase-based system to monitor *Trypanosoma cruzi* infection in mice by bioluminescence imaging. PLOS ONE. 2018; 13(4): e0195879.
- Calvet CM, Silva TA, Thomas D, Suzuki B, Hirata K, Siqueira-Neto JL, McKerrow JH. Long term followup of *Trypanosoma cruzi* infection and Chagas disease manifestations in mice treated with benznidazole or posaconazole. PLoS Neglected Tropical Diseases. 2020; 14(9):e0008726.
- de Castro Côbo E, Silveira TP, Micheletti AM, Crema E, Adad SJ. Research on *Trypanosoma cruzi* and Analysis of Inflammatory Infiltrate in Esophagus and Colon from Chronic Chagasic Patients with and without Mega. Journal of Tropical Medicine. 2012:232646.
- Adad SJ, Andrade DCdS, Lopes ER, Chapadeiro E. Contribuição ao estudo da anatomia patológica do megaesôfago chagásico. Rev I Med Trop. 1991; 33:443–50.
- 54. da Silveira ABM, Arantes RME, Vago AR, Lemos EM, Adad SJ, Correa-Oliveira R, D'Avila Reis D. Comparative study of the presence of *Trypanosoma cruzi* kDNA, inflammation and denervation in chagasic patients with and without megaesophagus. Parasitology. 2005; 131(05):627–34.
- 55. Pinto L, Schijman AG, Alonso-Padilla J, Lozano D, Torrico MC, Gamba P, et al. Molecular detection and parasite load of *Trypanosoma cruzi* in digestive tract tissue of Chagas disease patients affected by megacolon. Acta Trop. 2022; 235:106632. https://doi.org/10.1016/j.actatropica.2022.106632 PMID: 35932843
- 56. Rangel-Gamboa L, López-García L, Moreno-Sánchez F, Hoyo-Ulloa I, Vega-Mémije ME, Mendoza-Bazán N, et al. *Trypanosoma cruzi* infection associated with atypical clinical manifestation during the acute phase of the Chagas disease. Parasites & Vectors. 2019; 12(1):506.
- Riganti J, Maqueda MG, Piñero MCB, Volonteri VI, Galimberti RL. Reactivation of Chagas' disease: cutaneous manifestations in two immunosuppressed patients. International Journal of Dermatology. 2012; 51(7):829–34. https://doi.org/10.1111/j.1365-4632.2011.05224.x PMID: 22715827
- Sartori AM, Sotto MN, Braz LM, Oliveira Júnior Oda C, Patzina RA, Barone AA, Shikanai-Yasuda MA. Reactivation of Chagas disease manifested by skin lesions in a patient with AIDS. Transactions of the Royal Society of Tropical Medicine and Hygiene. 1999; 93(6):631–2. https://doi.org/10.1016/s0035-9203(99)90077-9 PMID: 10717752

- Alfituri OA, Quintana JF, MacLeod A, Garside P, Benson RA, Brewer JM, et al. To the Skin and Beyond: The Immune Response to African Trypanosomes as They Enter and Exit the Vertebrate Host. Front Immunol. 2020; 11:1250. https://doi.org/10.3389/fimmu.2020.01250 PMID: 32595652
- 60. Singh OP, Tiwary P, Kushwaha AK, Singh SK, Singh DK, Lawyer P, et al. Xenodiagnosis to evaluate the infectiousness of humans to sandflies in an area endemic for visceral leishmaniasis in Bihar, India: a transmission-dynamics study. Lancet Microbe. 2021; 2(1):e23–e31. <u>https://doi.org/10.1016/S2666-5247(20)30166-X PMID: 33615281</u>
- Morillo CA, Marin-Neto JA, Avezum A, Sosa-Estani S, Rassi AJ, Rosas F, et al. Randomized Trial of Benznidazole for Chronic Chagas' Cardiomyopathy. New England Journal of Medicine. 2015; 373 (14):1295–306. https://doi.org/10.1056/NEJMoa1507574 PMID: 26323937
- Stewart GL, Giannini SH. Sarcocystis, Trypanosoma, Toxoplasma, Brugia, Ancylostoma, and Trichinella spp.: A review of the intracellular parasites of striated muscle. Experimental Parasitology. 1982; 53(3):406–47. https://doi.org/10.1016/0014-4894(82)90083-2 PMID: 6806117
- Weaver JD, Hoffman VJ, Roffe E, Murphy PM. Low-level parasite Persistence drives vasculitis and myositis in skeletal muscle of mice chronically infected with *Trypanosoma cruzi*. Infection and Immunity. 2019; 87(6):e00081–19.
- Roffê E, Marino APMP, Weaver J, Wan W, de Araújo FF, Hoffman V, et al. *Trypanosoma cruzi* Causes Paralyzing Systemic Necrotizing Vasculitis Driven by Pathogen-Specific Type I Immunity in Mice. Infection and immunity. 2016; 84(4):1123–36.
- 65. Bombeiro AL, Gonçalves LA, Penha-Gonçalves C, Marinho CRF, D'Império Lima MR, Chadi G, Álvarez JM. IL-12p40 Deficiency Leads to Uncontrolled *Trypanosoma cruzi* Dissemination in the Spinal Cord Resulting in Neuronal Death and Motor Dysfunction. PLOS ONE. 2012; 7(11):e49022.
- Clowry GJ, Basuodan R, Chan F. What are the Best Animal Models for Testing Early Intervention in Cerebral Palsy? Front Neurol. 2014; 5:258. https://doi.org/10.3389/fneur.2014.00258 PMID: 25538677
- Genovese O, Ballario C, Storino R, Segura E, Sica RE. Clinical manifestations of peripheral nervous system involvement in human chronic Chagas disease. Arq Neuropsiquiatr. 1996; 54(2):190–6. https://doi.org/10.1590/s0004-282x199600020002 PMID: 8984973
- Sica RE, Genovese OM, Gargia Erro M. Peripheral motor nerve conduction studies in patients with chronic Chagas' disease. Arq Neuropsiquiatr. 1991; 49(4):405–8. <u>https://doi.org/10.1590/s0004-</u> 282x1991000400006 PMID: 1668781
- Laguens R, Cossio P, Diez C, Segal A, Vasquez C, Kreutzer E, et al. Immunopathologic and morphologic studies of skeletal muscle in Chagas' disease. The American Journal of Pathology. 1975; 80 (1):153. PMID: 125546
- 70. Torres SH, Finol HJ, Montes de Oca M, Vásquez F, Puigbó JJ, Loyo JG. Capillary damage in skeletal muscle in advanced Chagas' disease patients. Parasitol Res. 2004; 93(5):364–8. <u>https://doi.org/10.1007/s00436-004-1107-7 PMID: 15205941</u>
- Montes de Oca M, Torres SH, Loyo JG, Vazquez F, Hernández N, Anchustegui B, Puigbó JJ. Exercise performance and skeletal muscles in patients with advanced Chagas disease. Chest. 2004; 125 (4):1306–14. https://doi.org/10.1378/chest.125.4.1306 PMID: 15078739
- 72. Taratuto A, Pagano MA, Fumo T, Sanz OP, Sica RE. Histological and histochemical changes of the skeletal muscle in human chronic Chagas' disease. Arq Neuropsiquiatr. 1978; 36(4):327–31. <u>https:// doi.org/10.1590/s0004-282x1978000400006</u> PMID: 87186
- Benavente OR, Patiño OL, Peña LB, Lugònes H, Kalala E, Meneclier CR, et al. Motor unit involvement in human acute Chagas' disease. Arq Neuropsiquiatr. 1989; 47(3):283–6. https://doi.org/10.1590/ s0004-282x1989000300005 PMID: 2559681
- 74. Pinazo M-J, Espinosa G, Cortes-Lletget C, Posada EdJ, Aldasoro E, Oliveira I, et al. Immunosuppression and Chagas Disease: A Management Challenge. PLoS Neglected Tropical Diseases. 2013; 7(1): e1965. https://doi.org/10.1371/journal.pntd.0001965 PMID: 23349998